Author's response to reviews

Title: Successful MDR-TB treatment regimens including Amikacin are associated with high rates of hearing loss

Authors:

Chawangwa Modongo (ntungwana@yahoo.co.uk)
Rafal Sobota (rafal4484@gmail.com)
Boikobo Kesenogile (kesenogile@yahoo.co.uk)
Ronald Ncube (rtncube@gmail.com)
Giorgio Sirugo (gsirugo@gmail.com)
Scott Williams (Scott.Williams@dartmouth.edu)
Nicola M Zetola (nzetola@gmail.com)

Version: 2
Date: 4 February 2014

Author's response to reviews:

Adding emails of the co-authors